Author’s response to reviews

Title: Bacterial isolates and their antimicrobial susceptibility pattern among patients with external ocular infections at Borumeda hospital, Northeast Ethiopia

Authors:

Birtukan Shiferaw (birtukan_sheferaw@yahoo.com)
Baye Gelaw (tedybayegelaw@gmail.com)
Abate Assefa (abezew@gmail.com)
Yared Assefa (assefabi2003@yahoo.com)
Zelalem Addis (zelalemaddis@ymail.com)

Version: 4 Date: 24 April 2015

Author’s response to reviews:

Consent for illiterate participant

Dear editors,

We have included the ethical issue related to illiterate study participants as “For illiterate participants the written consent was read for them and they were included in the study when they explained their agreement verbally”. This statement is found on line 141-142 of the revised manuscript.